Author’s response to reviews

Title: Primary percutaneous coronary intervention during ST elevation myocardial infarction in prosthetic valve endocarditis: a case report

Authors:

Alfonso Campanile (alfonsocampanile@hotmail.it)
Guido Tavazzi (gtavazzi@yahoo.it)
Francesco Caprioglio (capriogliofrancesco@gmail.com)
Fausto Rigo (fausto.Rigo@aulss3.veneto.it)

Version: 2 Date: 16 Jan 2018

Author’s response to reviews:

Dear Editor,

Here is our reply to you and reviewer 2.

We have modified the paper as suggested by you and, in particular, in relation to the point 6 (Videos), we have removed the title from each clip and provided a detailed description and a descriptive title for each video in the Legend. We really hope that, now, the paper will be suitable for publication in BMC Cardiovascular Disorders.

Herewith we also wish, again, to thank you and all the reviewers for the positive comments and the essential contribute made to our work.

Kind Regards

Dr Alfonso Campanile
Regarding the comments of the Reviewer 2:

We have made all the further modifications suggested to the English style and we wish to thank, again, the Reviewer for his important support in improving the text.